Author’s response to reviews

Title: A school-based intervention improved dietary intake outcomes and reduced waist circumference in adolescents: a cluster randomized controlled trial.

Authors:

Angelica Ochoa (angelica.ochoa@ucuenca.edu.ec)

Rosmarijn Verstraeten (roosmarijn.verstraeten@gmail.com)

Lieven Huybregts (l.huybregts@cgiar.org)

Silvana Donoso (silvana.donosom@ucuenca.edu.ec)

Susana Andrade (susana.andrade@ucuenca.edu.ec)

John Van Camp (john.vancamp@ugent.be)

Patricia Ramírez (patricia.ramirez@ucuenca.edu.ec)

Carl Lachat (carl.lachat@ugent.be)

Lea Maes (lea.maes@ugent.be)

Patrick Kolsteren (patrick.kolsteren@ugent.be)

Version: 3 Date: 21 Nov 2017

Author’s response to reviews:

Changes are highlighted in yellow in the main manuscript.

- In the abstract: adolescents’ was corrected

- Line 419 now reads: We acknowledge a number of limitations. The 24-hour recalls were applied only during weekdays